Author's response to reviews

**Title:** Urinary High Molecular Weight Matrix Metalloproteinases as non-invasive biomarker for detection of bladder cancer

**Authors:**

Mohammed A Mohammed (mamnci@yahoo.com)
Manar F Seleim (m_fouda80@yahoo.com)
Mohga S Abdalla (drmohgashafik@yahoo.com)
Hayat M Sharada (hayatsharadamk@gmail.com)
Abdel Hady A Abdel Wahab (abdelhadya@gmail.com)

**Version:** 2  **Date:** 22 November 2012

**Author's response to reviews:** see over
Dear Editor,

We herein submit our original contribution entitled “Urinary High Molecular Weight Matrix Metalloproteinases as non-invasive biomarker for detection of bladder cancer” to BMC Urology.

We certify that our manuscript represents original work not previously published in any substantial part and is not under consideration of publication elsewhere. All authors have no conflict of interest to disclose. All authors certify that they have read and approved this manuscript for submission, have met the criteria for authorship as established by the International Committee of Medical Journal Editors, believe that the paper represents honest work, and are able to verify the validity of the results reported. We also certify that in case of acceptance of the manuscript the copyright is transferred to BMC Urology.

Sincerely,

Abdel Hady A. Abdel Wahab, PhD.

Mailing Address: 1 Kasr El-Aini Str. Department of Cancer Biology, National Cancer Institute, Cairo University, Cairo, Egypt.